Author’s response to reviews

Title: Standard (8 weeks) vs long (12 weeks) Timing to Minimally-Invasive Surgery after NeoAdjuvant Chemoradiotherapy for Rectal cancer: a multicenter randomized controlled parallel group trial (TiMiSNAR). Protocol paper.

Authors:

Igor Monsellato (igor.monsellato@ospedale.al.it)
F. Alongi (filippo.alongi@sacrocuore.it)
E. Bertocchi (elisa.bertocchi@sacrocuore.it)
S Gori (stefania.gori@sacrocuore.it)
G. Ruffo (giacomo.ruffo@sacrocuore.it)
E. Cassinotti (cassinotti.elisa@gmail.com)
L. Baldarti (ludovica.baldarti@gmail.com)
L. Boni (luigi.boni@policlinico.mi.it)
G. Pernazza (gpernazza@hsangiovanni.roma.it)
F. Pulighe (fabiopulighe@gmail.com)
C. De Nisco (carlodenisco@gmail.com)
R. Perinotti (roberto.perinotti@aslbi.piemonte.it)
E. Morpurgo (emilio.morpurgo@aulss6.veneto.it)
T. Contardo (tania.contardo@aulss6.veneto.it)
E. Mammano (enzo.mammano@aulss6.veneto.it)
U. Elmore (elmore.ugo@hsr.it)
R Delpini (delpini.roberto@hsr.it)
R Rosati (rosati.riccardo@unisr.it)
F. Perna (fedefez@hotmail.com)
Title: Standard (8 weeks) vs long (12 weeks) Timing to Minimally-Invasive Surgery after NeoAdjuvant Chemoradiotherapy for Rectal cancer: a multicenter randomized controlled parallel group trial (TiMiSNAR). Protocol paper.

Authors:

Igor Monsellato
Filippo Alongi
Elisa Bertocchi
Stefania Gori
Giacomo Ruffo
Elisa Cassinotti
Ludovica Baldarti
Luigi Boni
Graziano Pernazza
Fabio Pulighe
Carlo De Nisco
Roberto Perinotti
Emilio Morpurgo
Tania Contardo
Enzo Mammano
Ugo Elmore
Roberto Delpini
Riccardo Rosati
Federico Perna
Andrea Coratti
Benedetta Menegatti
Sergio Gentilli
Paolo Baroffio
Piero Buccianti
Riccardo Balestri
Cristina Ceccarelli
Valter Torri
Elena Traverso
Vittorio Fusco
Maura Rossi
Fabio Priora
Gianmauro Numico
Paola Franzone

Version: 3 Date: 28/09/2019

Author’s response to Editor’s Comments:

1. Authorship change form

-- We note that the order of authors has changed since original submission of the manuscript. In line with COPE guidelines, BioMed Central requires written confirmation from all authors that they agree with any proposed changes in authorship of submitted manuscripts or published articles. In such cases, we use a standardised form which we would request you and your co-authors to complete. The authorship change form can be found at the following link: https://resource-cms.springernature.com/springer-cms/rest/v1/content/7454878/data/v5. All instructions can be found on the form, please treat the ‘current authorship’ section as the original authorship. Please return the form by email to the Editorial Office (Or include this a supplementary file in your revised manuscript) with all author signatures (including those newly added/removed).

R: The Authorship form has been updated with the signatures all the authors.

2. Author initials

-- We note that several pairs of authors have the same initials - for instance, EM and FP. To distinguish between them in the Authors’ Contributions section please designate them as follows in the example used: EM1 and EM2, with EM1 corresponding to E. Morpugo, chosen because this name is furthest up the author list.

R: Initials have been fixed and updated as requested

3. Author contributions
Currently, we note several discrepancies in the "Authors Contributions" statement in your Declarations section. Author initials MS is included, though we do not note an author with the initials MS listed in the most recent version of the manuscript. Further, authors PB2, RB, CC, VF, MR and FP3 are currently missing from the statement, despite being listed in the most recent version of the manuscript. Please ensure all authors listed as having contributed to the manuscript are featured in adequate and accurate detail in this section.

R: All issues have been fixed

4. Abbreviations

-- Please provide a list of all the abbreviations used in the manuscript. This list should be placed just before the Declarations section. All abbreviations should still be defined in the text at first use.

R: All abbreviations have been reported and updated and clearly specified in the text

5. Acknowledgements

-- We note that you have not included an acknowledgements section. If you have no acknowledgements please put ‘Not Applicable’ in this section. Please do not include any funding information in your acknowledgements.

R: an Acknowledgment section has been added as requested

6. Clean manuscript

-- At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

R: A clean copy of the manuscript has been uploaded

A new authorship form has been uploaded